Author's response to reviews

Title: 3D-Ultrasonography for Evaluation of Facial Muscles in Patients with Chronic Facial Palsy or Defective Healing: a Pilot Study

Authors:

Gerd F Volk (fabian.volk@med.uni-jena.de)
Martin Pohlmann (martin.pohlmann87@web.de)
Mira Finkensieper (mira.finkensieper@med.uni-jena.de)
Heather J Chalmers (heather.chalmers@uoguelph.ca)
Orlando Guntinas-Lichius (orlando.guntinas@med.uni-jena.de)

Version: 4
Date: 11 April 2014

Author's response to reviews: see over
Response to the reviewers:

Dear Sirs:

Thank you to the referee and his comments!

1. Referee #2:
The manuscript became suitable for publication.
Answer: No response necessary.

3. Editorial requests:
Please include full contact information, including emails, for all authors on the title page.
Answer: Done.

Orlando Guntinas-Lichius
For all authors
Jena, 11-April-2014